Author’s response to reviews

Title: Differences in Gram-positive bacterial colonization and antimicrobial resistance among children in a high income inequality setting

Authors:

Felipe Neves (fpgneves@id.uff.br)
Mariel Marlow (klt8@cdc.gov)
Gabriel Rezende-Pereira (rezendepgabriel@gmail.com)
Marcos Pinheiro (marcosgabrielp@gmail.com)
Allyne Santos (allyne_1993@hotmail.com)
Maria de Fátima Freitas (nogueiradefreitasalves@gmail.com)
Rosana Barros (rrbarros@id.uff.br)
Fábio Aguiar-Alves (faalves@gmail.com)
Claudete Cardoso (claudetecardoso@id.uff.br)
Lee Riley (lwriley@berkeley.edu)

Version: 2 Date: 14 May 2019

Author’s response to reviews:

Dear Sir,

We would like to thank the reviewers for their valuable comments and suggestions, which were helpful to improve the manuscript and make it clearer. Responses to specific editor comments are below.

We would like to thank the reviewers for their valuable comments and suggestions, which were helpful to improve the manuscript and make it clearer. Responses to specific editor comments are below.
Editor Comments:

1. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We included the following statement. "Funders have not contributed to the design of the study and collection, analysis, and interpretation of data and in writing the manuscript". (lines 319-320)

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Done